
MEG Management Committee meeting

23.03.2010, 14:00hrs, West Wing Seminar Room, CBU



Minutes






· Present:  Friedemann Pulvermuller, Rik Henson, Danny Mitchell, Yury Shtyrov, Olaf Hauk, Rob Anthony, Maarten Van Casteren, Clare Cook, Georgie Morrill.

1. Apologies for absences

· Gary Chandler
· Maarten Van Casteren

2. Matters arising 

RH informed the committee that he’d had problems with Maxfilter crashing and having to re-boot.  For now FP suggested that the issues and any ways of solving them should be put on the Wiki for other researchers to see.

FP enquired whether the empty room recordings were still taking place and MvC, CC and OH confirmed that they carry these out every morning after tuning the machine.
Action: 
MEG committee members to put ‘trouble-shooting’ for Maxfilter on the MEG Wiki.

3. Minutes of the last meeting

Approved.

4. Computing
							
RA reported that Maxfilter just needs to be made available to other systems at some point and suggests that he should add MCE interlinking with it also.  RH asked for confirmation that both versions of MaxFilter could not be installed on one machine, and RA informed that there are ways to do this but it becomes quite complicated.  The easiest way to do this is to have both versions centrally and then link to the nearest directory.  RH thinks that we’re still not in a position to advise which is the best version to use. YS said he would advise researchers to use the older version.  RH is concerned that having this on L58 could make it harder for more people to test.  RA suggested that that he could store it centrally and map it out to those machines so anything from L43 and above would have the new version and the new suite.
Action: RA to store the old and new versions of Maxfilter centrally and map out to the relevant machines.

MvC informed the committee that that he unfortunately infected the stimulus computer with a virus from his Memory stick and only found out because of using the stick in another computer which had the virus scanning.  The Stimulus computer isn’t on the network so has no up to date virus software.  RA informed that the reason it wasn’t put on the network was so that virus’s wouldn’t spread to all machines.  There are a few plans in place such as people using encrypted memory sticks which when inserted in the computer will bring up a virus scan automatically.
Action: RA to liaise with GC.

5. Lab business and progress

5.1.  Lab re-opening update
YS reported that the wall for sound insulation and annual service went to plan.  CC and MvC confirmed there had been no issues to date. 
5.2. Final payment to Elekta (for wall installation)
The committee recommended payment in full to Elekta.
5.3. Eye Tracker upgrade
MvC informed the committee that the Eye Tracker is still faulty.  Elekta have agreed to fix in the short term and then will upgrade as soon as the newer version is available.
Action: MvC to chase if no response from Elekta.
5.4. Other Issues
OH brought up the issue of the wrong MEG forms being on the MRI admin page.  They are named “MEG consent Form” and “MEG information Sheet”, but both are fMRI not MEG.
Action: GM to look into and change
OH suggested it would be good to share data on the Wiki.  FP agreed and suggested having a list of topics relating to MEG software support.
Action: OH to put on Wiki. 

6. Support request for Gold Standard project 

RH and OH asked for support in the "gold standard" project. This project will be relevant for methods evaluation and teaching at the CBU, and so support in volunteer booking and data acquisition within the next few months, e.g. from the MEG operators, would be a big help. The aim is to use 12 subjects, EEG+MEG as well as structural MRI. Operators agreed they would be happy to help with lab support during quiet periods in the lab, and OH added that the students can be responsible for other aspects like the booking of subjects etc. 
Action: CC/MvC to help with lab support and OH to speak to students about helping out with participant booking etc.



7. MEG request form

The committee looked at the current and proposed form and decided that it was pretty much the same but in a different format.
Action: GM to amend and put on Wiki and remind researchers to use the new form.

8. MEG lab stats for February

In February, the MEG lab was operational 62 hours, with 114 hours available for experimental studies amounting to 54 percent used time.
	Action: GM to include MEGMC in email to announce available slots to researchers.
9. AOB

RH informed the committee of the LoLa project which had been awarded £5million from a BBSRC LoLa grant to investigate the cognitive neuroscience of ageing. The plan is to record 700 participants over 5 years (probably starting around January 2011).  They will require 4 x 1 hour slots per week (they will have fMRI scans too) and RH asked if these slots would be able to be booked in advance.  The committee agreed that this shouldn’t be a problem but would look into it nearer the time.

Date of the next meeting

Wednesday 24th March 2010


 

